RESEARCH

Open Access

Inpatient hospital course and self-reported symptomatology in underweight adults with ARFID compared to age- and sex-matched controls with anorexia nervosa

Irina A. Vanzhula^{1*}, Erin Wang¹, Mary K. Martinelli¹, Colleen Schreyer¹ and Angela S. Guarda¹

Abstract

Objective Avoidant restrictive food intake disorder (ARFID) has similar prevalence to anorexia nervosa (AN) in adults, but research in this population is lacking. Although inpatient or residential treatment involving nutritional rehabilitation is increasingly recommended for malnourished individuals with ARFID, best practices remain poorly defined. Existing studies on self-reported symptomatology and treatment course and outcome are primarily in child and adolescent cohorts and demonstrate inconsistent findings. This study aimed to compare hospital course and self-reported symptomatology of underweight adult inpatients with ARFID and sex- and age-matched patients with AN.

Method Underweight adult patients with ARFID or AN admitted to a specialized, hospital-based behavioral treatment program completed measures of body dissatisfaction, drive for thinness, bulimic symptoms, anxiety, depression, and personality traits. Demographic and treatment course data were abstracted from electronic medical records. Patients with ARFID (n=69) were matched to those with AN (n=69) based on sex and age.

Results Adults with ARFID were closer to target weight at admission, but gained weight at a slower rate, were discharged at lower BMI, and were less likely to reach target weight by discharge than adults with AN. Patients with ARFID reported less weight and shape-related eating disorder, state anxiety, and depression symptoms and lower neuroticism.

Discussion Adults with ARFID progress through treatment more slowly and achieve less favorable weight outcomes by hospital discharge than patients with AN, but long-term outcomes are unclear. Describing clinical presentations and course of illness of adult ARFID may help inform treatment protocols.

Keywords Anorexia nervosa, ARFID, Inpatient treatment, Weight restoration

Plain English Summary

This study examined hospital course and symptomatology in underweight adults with avoidant/restrictive food intake disorder (ARFID) compared to adults with anorexia nervosa (AN). Both groups were admitted at similar BMI,

*Correspondence: Irina A. Vanzhula ivanzhu1@jhu.edu ¹Johns Hopkins University School of Medicine, Meyer 101, 600 N. Wolfe Street, Baltimore, MD 21287, USA



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/ficenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

however compared to patients with AN, those with ARFID gained weight slower and were half as likely to reach target weight by discharge. Adults with ARFID were equally likely, however, to be rated as clinically improved at discharge. Patients with ARFID also reported less symptoms of anxiety, depression or neuroticism than did those with AN, and less weight and shape-related eating disorder symptoms at admission. The reason for slower weight restoration in adults with ARFID may reflect the need for more individualized protocols adapted to meet the unique, often heterogeneous needs of these individuals. Longer-term post-discharge outcomes for adults with ARFID remain unclear and require investigation.

Introduction

Avoidant/restrictive food intake disorder (ARFID) is defined as a persistent disturbance in feeding or eating that results in severe malnutrition, significant weight loss or failure to gain weight, or growth compromise that impairs psychosocial functioning [1]. In ARFID, restrictive eating behaviors are associated with apparent lack of interest in eating, avoidance based on the sensory characteristics of food, or concerns about the aversive consequences of eating. Notably, eating behaviors in ARFID are not primarily motivated by weight or body image concerns [1]. ARFID commonly begins in early childhood and, if untreated, may persist into adulthood [14]. Point prevalence of full threshold ARFID based on selfreported symptoms was estimated at 0.8% for women and 0.9% for men [14]. Nakai et al. [22] found that ARFID accounted for 9.2% of all eating disorder (ED) hospital admissions (mixed adult and adolescent sample). Similar to anorexia nervosa (AN), patients with ARFID may experience medical complications and sometimes require hospitalization for nutritional rehabilitation [15, 22], but best practices for such interventions are yet to be defined. More research is needed to describe clinical presentations and illness course of adult ARFID to inform treatment guidelines.

Few studies examined ARFID symptomatology in adults. Zickgraf et al. [34] found that adult participants with ARFID symptoms endorsed high levels of quality of life impairment, internalizing symptoms and obsessivecompulsive symptoms. Another study observed adults with self-reported ARFID symptoms endorsed lower weight and shape-related ED symptomatology but similar levels of anxiety and depression as participants with other EDs,ARFID symptoms were unrelated to age and sex [14]. Both Thomas et al. [33] and Manwaring et al. [20] reported clinical levels of anxiety and depression in adults with ARFID. Fjeldstad et al. [8] compared adults with ARFID to those with AN and found lower selfreported anxiety symptoms and lower weight and shaperelated ED symptomatology in adults with ARFID, but no differences in depression. Several other studies used mixed-age samples (adolescents and adults) to characterize ARFID. For example, Nakai et al. [22] found that individuals with ARFID scored lower than those with AN on drive for thinness, bulimic symptoms, body dissatisfaction, perfectionism, and interpersonal distrust. Becker et al. [3] also found that individuals with ARFID scored lower than those with AN on most measures of weight and shape-related psychopathology, except for restrictive eating. Although both anxiety and depression were lower in ARFID than AN, mean anxiety scores were within a clinical range [3]. Overall, it appears that individuals with ARFID experience similar impairment to those with other EDs, although more research in adults is needed.

Due to poor nutritional intake or dangerously low weight, patients with ARFID or AN may require hospitalization for normalization of eating behaviors and weight restoration. The majority of studies examining ARFID hospital course were in adolescent or mixed-age samples. Several studies of hospitalized patients with ARFID have found that adolescents and young adults with ARFID have similar body mass index (BMI) or percent target weight at hospital admission compared to that of patients with AN [8] (all adult sample); [18, 22, 30], but one study reported higher admission BMI in adolescents and young adults with ARFID [19]. The findings regarding length of stay are inconsistent, with one study reporting longer length of stay for a mixed age sample with ARFID versus AN [30] and another reporting no difference in length of stay [19]. Although one study observed a slower rate of weight gain in youth with ARFID compared with that of AN (1.36 kg vs. 1.92 kg/week respectively; [19]), another found no difference [30]. Adolescents and young adults with ARFID and AN showed comparable weight outcomes at hospital discharge (e.g., discharge BMI, % discharged for clinical improvement [18, 19]. Two studies have examined longer term outcomes post discharge, with one study finding that adolescents with ARFID had a higher rate of recovery at follow-up than did those with AN (77% ARFID and 43% AN recovered at 2-37 months post-discharge) [18]. The second study found similar remission rates at 1-year follow up (62% ARIFD and 46% AN remitted) although this difference did not reach statistical significance [32].

Overall, findings regarding hospital course and outcomes of patients with ARFID vs AN are inconsistent, which is likely due in part to very low sample sizes for the ARFID group in most of these studies (N range=7–41). The sample sizes of AN and ARFID groups were also uneven, with the AN sample often being several times larger. Both these issues can introduce bias in the analyses. Further, all the studies of hospital course except for one [8] focused on children and adolescents or had a mixed age sample. More research on treatment course and outcomes in hospitalized adult patients with ARFID using larger samples is needed to inform treatment protocols and approaches to renourishment in patients hospitalized with underweight ARFID. This study compared inpatient treatment course of underweight adult patients with ARFID and sex- and age-matched patients with AN (total N = 138; 69 AN and 69 ARFID). To minimize sample bias, we performed case control matching based on sex and age to obtain samples similar in demographic characteristics. Because of inconsistent findings in the prior literature regarding treatment outcomes, no hypotheses were generated. We also compared selfreported symptomatology of patients with ARFID vs AN in a subset of the sample that had this data available (total N=54; 27 AN and 27 ARFID). Despite the smaller sample size, these analyses were included due to the paucity of literature on adults with ARFID, although associated findings are preliminary and should be interpreted with caution. We hypothesized that patients with ARFID would report lower ED and depression symptoms than patients with AN. We did not make hypotheses regarding anxiety, desired weight, or personality traits because of conflicting findings or lack of prior research.

Methods

Participants

The sample consisted of 138 hospitalized underweight adult patients with ARFID or AN admitted to a specialized behavioral inpatient treatment program for eating disorders between 2003 and 2022. Patients with ARFID (n=69) were matched to those with AN (n=69) based on sex and age. A subsample of adults with ARFID (n=27) who completed self-report questionnaires at admission were matched to 27 patients with AN based on sex and age (total sample with available self-report data N=54). Sample demographics are presented in Table 1.

Procedures

Clinical and demographic data were collected as part of routine clinical care. A retrospective chart review approved by the Institutional Review Board of the Johns Hopkins University School of Medicine allowed

Ta	bl	e	1	Samp	le c	lemo	graph	ic ar	nd d	descri	iptive	stat	istics	chara	cteristic	S
----	----	---	---	------	------	------	-------	-------	------	--------	--------	------	--------	-------	-----------	---

	Full Sample (N = 138)		Subsample ^a (<i>n</i> = 54)	
	ARFID (<i>n</i> =69)	AN (<i>n</i> =69)	ARFID (<i>n</i> = 27)	AN (n=27)
Age, M (SD)	35.86 (16.45)	35.86 (16.45)	32.96 (14.01)	32.96 (14.01)
BMI at admission, M (SD)	16.38 (1.85)	15.86 (1.91)	15.99 (1.55)	15.65 (1.90)
Length of stay (days), M (<i>SD</i>)	40.39 (26.42)	59.46 (31.11)	47.41 (22.75)	55.26 (27.35)
Sex, n (%)				
Female	61 (88.4%)	61 (88.4%)	23 (85.2%)	23 (85.2%)
Male	8 (11.6%)	8 (11.6%)	4 (14.8%)	4 (14.8%)
Ethnicity, n (%)				
Hispanic or latino	3 (4.3%)	0 (0%)	0 (0%)	1 (3.7%)
Not hispanic or latino	65 (94.2%)	67 (97.1%)	25 (92.6%)	25 (92.6%)
Chose not to disclose	1 (1.4%)	2 (2.9%)	2 (7.4%)	1 (3.7%)
Race, n (%)				
American Indian/Alaskan Native	0 (0%)	0 (0%)	0 (0%)	0 (0%)
Asian	1 (1.4%)	2 (2.9%)	0 (0%)	1 (3.7%)
Native Hawaiian or Pacific Islander	0 (0%)	0 (0%)	0 (0%)	0 (0%)
Black or African American	7 (10.1%)	0 (0%)	0 (0%)	2 (7.4%)
White or Caucasian	57 (82.6%)	66 (95.7%)	26 (96.3%)	22 (81.5%)
Multiracial	2 (2.9%)	0 (0%)	0 (0%)	1 (3.7%)
Other	1 (1.4%)	0 (0%)	0 (0%)	0 (0%)
Chose not to disclose	1 (1.4%)	1 (1.4%)	1 (3.7%)	1 (3.7%)

BMI, body mass index (kg/m²); %TW, percent target weight; AN, anorexia nervosa; ARFID, avoidant/restrictive food intake disorder. ^aA subset of patients from the full sample consented to complete additional self-report questionnaires at admission

abstraction of de-identified demographic and clinical data from the electronic medical record. This data included admission height and weight, pounds from target weight at admission, weight at discharge, length of stay (in days), achievement of target weight (yes or no), and reason for discharge (for clinical improvement vs not for clinical improvement [e.g., against medical advice, insurance, etc.]). Additional self-reported questionnaire data were collected from a subset of these patients who consented to participate in a longitudinal outcomes research study. Questionnaire data were collected using paper and pencil within the first week of admission. ED diagnoses were determined via the Structured Clinical Interview for the Diagnostic and Statistical Manual of Mental Disorders, 5th Edition (DSM-5; [7]). Diagnostic interviews were administered by postdoctoral fellows or trained research assistants supervised by a licensed clinical psychologist. Prior to the existence of DSM-5 and the introduction of the ARFID diagnosis, this study would have classified ARFID cases as Eating Disorder Not Otherwise Specified. After the release of DSM-5, all diagnoses were reviewed by a team of psychiatrists and clinical psychologists to ensure diagnoses consistently reflected the revised DSM-5 criteria. The SCID was administered by only one rater, and therefore inter-rater reliability is not available.

Treatment protocol

The inpatient eating disorders program employs a structured meal-based behavioral treatment protocol delivered within a multidisciplinary integrated, inpatient program. Primary treatment goals include rapid weight restoration for underweight patients and normalization of eating behavior. The nutritional protocol is 100% meal-based and nasogastric feeds are never employed. Patients consume three supervised meals a day in a group setting. Calories are advanced from 1200 to 2000 kcals/ day (depending on admission BMI) to 3500-4000 kcals/ day for individuals on weight gain protocol [11]. Calories above 2500 are administered via snacks and liquid supplements. With rare exceptions (e.g., patient was vegetarian at least three years prior to developing an ED, religious exceptions, or documented food allergy), food preferences or dislikes are not accommodated, as the goal is to help patients diversify their food intake. Patients may begin selecting menu items contingent upon completing 100% of meals served once at a calorie intake level of 3500 cal/day typically by day 10–12. Menu selections are reviewed by staff for compliance with the exchange plan. The nutritional rehabilitation protocol was identical for patients with ARFID and AN.

Measures Weight and BMI

Height and weight measured at admission and discharge were used to compute admission and discharge BMI. Individualized target weights set for each patient were a four-pound range (1.8 kg) based on the patient's age, sex, and height centered on a BMI of 20.5 kg/m² for patients over age 25. For patients aged 18-24, target weight was determined using growth charts when available. For patients whose baseline BMI was above the 50th percentile target BMI was set at the 50th BMI percentile. For those whose baseline BMI curve was below the 50th percentile but above the 25th, target BMI was calculated to fall on their premorbid BMI curve and for patients whose baseline BMI trajectory was below the 25th percentile, target BMI was set to the 25th BMI percentile. When growth charts were unavailable, target weight was set using the formula for adults over the age of 25 and adjusted by subtracting one pound (0.45 kg) per year of age below 25. Desired weight was assessed as a part of self-report questionnaires with one item ("How much would you like to weigh?").

Eating disorder inventory-2 (EDI-2)

The EDI-2 is a 64-item self-report measure of cognitive and behavioral characteristics of weight and shaperelated eating disorders [9]. Three subscales were used in the current study: Drive for Thinness, Bulimia, and Body Dissatisfaction. The EDI-2 has demonstrated good validity and reliability [9, 32] in patients with AN and BN, but no data are available on psychometric properties of the EDI-2 in patients with ARFID. Internal consistency for Cronbach's alphas for Drive for Thinness, Bulimia, and Body Dissatisfaction were excellent in this study (α =0.94, α =0.83, and α =0.91).

Eating disorder recovery self-efficacy questionnaire (EDSRQ)

The EDSRQ is a 23-item self-report measure of selfefficacy to refrain from acting on eating disordered behaviors and attitudes [21]. The Normative Eating Self-Efficacy subscale assesses confidence to eat without engaging in disordered eating behaviors (e.g., restricting, binge eating) and the Body Image Self-Efficacy subscale assesses confidence to maintain a realistic body image not dominated by pursuit of thinness. The EDRSQ has demonstrated good validity and reliability among individuals with weight and shape-related eating disorders [21, 24], but no psychometric data is available in ARFID samples. Internal consistencies for Normative Eating Self-efficacy and Body Image Self-efficacy were excellent in this study (α =0.97, α =0.93).

State and trait anxiety inventory (STAI)

The STAI is a 40-item self-report measure of anxiety as experienced in the moment (state subscale; STAI-S) and as a stable personality trait (trait subscale; STAI-T; [27]). The STAI has demonstrated good reliability and validity [12, 23]. Internal consistency for state and trait anxiety were excellent in the current study (α =0.97, α =0.94).

Beck depression inventory-II (BDI-II)

The BDI-II [2] is a 21-item self-report measure of depressive symptomatology. The BDI-II has strong psychometric properties, including internal consistency and factor validity [2, 28]. Internal consistency for the BDI-II in this study was excellent ($\alpha = 0.91$).

NEO five-factor inventory (NEO-FFI)

The NEO-FFI [5] is a widely-used 60-item self-report questionnaire used to provide a concise measure of the big five personality factors including neuroticism, extraversion, openness to experience, agreeableness, and conscientiousness. The NEO-FFI has demonstrated adequate psychometric properties among individuals with weight and shape-related eating disorders [31]. Internal consistencies in the current study were excellent for neuroticism (α =0.91), good for extraversion (α =0.80) and conscientiousness (α =0.87) and acceptable for openness to experience (α =0.76) and agreeableness (α =0.61).

Data analysis

Power analysis

According to G*Power [6], a sample size of n=64 in each group is required to detect a moderate effect size and a sample size of n=26 is required to detect a large effect using an independent t-test analysis (power=0.80, $\alpha = 0.05$).

Treatment course

All analyses were conducted in SPSS v 28. The total sample (N=138) was used for treatment course and outcomes analyses. Independent t-tests were conducted to compare patients with ARFID and AN on the following variables: Admission BMI, pounds from target weight at admission, length of stay, rate of weight gain, and discharge BMI. Pearson chi-square tests examined whether groups differed in whether they achieved target weight at discharge and reasons for discharge (clinical improvement vs not clinical improvement). Clinical improvement indicated that the patient had made sufficient progress in treatment and was ready to step down to a

lower level of care for continued treatment and weight restoration. Not for clinical improvement indicated that the patient was discharged for other reasons including financial concerns, patient/family request (against medical advice), administrative discharge for non-compliance, transfer to medical or other psychiatric unit, or elopement.

Self-reported symptomatology

The subsample of patients who completed self-report questionnaires (N=54) was used for these analyses. Independent t-tests were conducted to compare patients with ARFID and AN on the following admission variables: weight and shape-related ED symptoms, normative eating and body image self-efficacy, desired weight, anxiety, depression, and personality traits.

Results

Treatment course

Although patients with ARFID did not significantly differ from patients with AN on admission BMI, they were closer to their target weight at admission (Table 2). We conducted a post-hoc test comparing target weight of patients with ARFID and AN and found that for those under the age of 25, the ARFID group had lower target weights compared to AN (t(136) = 3.30, p = 0.002). No group differences in target weight were found for adults 25 and older with AN and ARFID (p=0.636). Groups did not significantly differ in length of stay. However, patients with ARFID gained weight at a significantly slower rate, had lower discharge BMI, and were less likely to reach target weight by discharge than patients with AN ($\chi 2[1, 138] = 9.44$, p = 0.002). Sixty percent of patients with AN and only 33% of patients with ARFID reached target weight before discharge. There were no differences between groups on reason for discharge (clinical improvement vs not clinical improvement; χ^2 [1, [138] = 0.26, p = 0.607). Fifty eight percent of patients with AN and 54% of patients with ARFID were discharged for clinical improvement. Out of 29 patients with AN discharged not for clinical improvement, 14 discharges were rated as patient or family-initiated, five were for financial reasons, five for non-compliance, three were transfers to other medical or psychiatric units, and two patients eloped. Out of 31 patients with ARFID discharged not for clinical improvement, 18 discharges were rated as patient or family-initiated, two were for financial reasons, five were for non-compliance, and six were transfers to other medical or psychiatric units.

Table 2 Comparison of patients with ARFID vs AN

Treatment course	ARFID (n = 69)	AN (n=69)		
	M (SD)	M (SD)	t	p
Admission BMI	16.38 (1.85)	15.86 (1.91)	- 1.63	.106
Percent target weight (Admission)	79.39(8.68)	77.19(9.75)	- 1.36	1.74
Pounds from target weight	-23.41 (11.42)	- 28.14 (12.54)	-2.31	.022
Rate of weight gain (lbs per week)	3.04 (2.02)	4.09 (1.35)	3.59	<.001
Length of stay (days)	31.43 (19.55)	34.51 (21.28)	.88	.379
Discharge BMI	19.14 (2.02)	20.01 (1.54)	2.79	.006
Admission characteristics	ARFID (n = 27)	AN (n=27)		
	M (SD)	M (SD)	t	Р
EDI-2 Drive for thinness	11.62 (3.48)	31.50 (10.45)	10.99	<.001
EDI-2 Bulimia	9.82 (3.74)	15.64 (8.46)	3.84	<.001
EDI-2 Body dissatisfaction	23.96 (6.52)	39.69 (1.79)	6.14	<.001
EDRSQ—Normative eating self-efficacy	4.05 (.86)	2.06 (1.10)	-7.76	<.001
EDRSQ—Body image self-efficacy	3.68 (.74)	1.91 (.91)	- 7.82	<.001
Desired BMI	20.46 (2.20)	17.41 (2.59)	-2.70	.010
STAI-state anxiety	43.36 (14.47)	61.40 (11.08)	2.58	.017
STAI-trait anxiety	44.47 (12.61)	56.80 (13.36)	1.92	.068
BDI-Depression	20.69 (10.87)	30.11 (12.93)	2.84	.006
NEO—Neuroticism	23.85 (7.96)	33.26 (11.01)	3.58	<.001
NEO—Extraversion	26.85 (8.18)	23.46 (6.91)	- 1.63	.110
NEO—Openness	31.15 (5.90)	28.03 (7.03)	- 1.75	.087
NEO—Agreeableness	33.37 (5.59)	31.42 (6.24)	- 1.19	.237
NEO—Conscientiousness	32.25 (8.49)	31.69 (7.63)	26	.800

ARFID, avoidant/restrictive food intake disorder; AN, anorexia nervosa; BMI, body mass index (kg/m²); EDI, Eating Disorder Inventory; EDSRQ, Eating Disorder Recovery Self-Efficacy Questionnaire; STAI, State-Trait Anxiety Inventory; BDI, Beck Depression Inventory

Self-reported symptomatology

Patients with ARFID scored lower on ED symptoms, state anxiety, depression, and neuroticism and higher on Normative Eating and Body Image Self-efficacy (Table 2) compared to those with AN. There were no differences between groups on trait anxiety or other personality variables (i.e., openness to experience, etc.). Patients with ARFID endorsed a higher desired BMI than patients with AN.

Discussion

Little is known about the presentation and treatment of ARFID in adults. The current study compared treatment course and self-reported symptomatology in underweight adults with ARFID admitted to an inpatient ED behavioral unit to age and sex-matched adults with AN. Adults with ARFID endorsed less psychopathology, including ED symptoms, state anxiety, depression, and neuroticism than patients with AN. Despite being closer to target weight, adults with ARFID gained weight at a slower rate, were discharged at lower BMI, and were less likely to reach target weight by discharge than adults with AN. While these results may suggest less favorable inpatient outcomes for adults with ARFID, patients in both groups were equally likely to be discharged for clinical improvement. Follow-up data is needed to evaluate short- and long-term outcomes of inpatient treatment for adults with ARFID.

Treatment course

Consistent with prior literature, we found that adults with ARFID had similar BMI at admission to those with AN (e.g., [8, 18, 22]). Adults with ARFID had less weight to gain to reach their target weight, which was explained by lower average goal weight ranges for ARFID compared to AN in those aged 18-25 years. Growth charts would have been incorporated into setting target weight for this age group if available. It is possible that a lower target weight was set for a subset of patients with ARFID who had a lifetime history of low weight. However, growth charts were not available to test this hypothesis. Steinberg et al. [29] also found that an individualized approach to setting target weight (using growth charts) resulted in lower target weight in children and adolescents with ARFID compared to other ED diagnoses. Further investigation on this issue is warranted. Like Makhzoumi et al.

[19], we found no differences in length of stay between adults with ARFID and AN, but patients with ARFID gained weight at a slower rate than those with AN. These similarities are not surprising considering both studies used data from patients admitted to the same treatment program (samples were partially overlapping).

While prior literature suggested similar discharge BMI for samples of adolescents and young adults with ARFID and AN [18, 19], we found that adults with ARFID had lower BMI at discharge and were less likely to achieve target weight by the end of their hospital stay than patients with AN. Only one third of ARFID patients reached target weight by discharge, compared with 60% of patients with AN. However, similar proportions were discharged for clinical improvement in both groups. Because patients with ARFID do not fear gaining weight, a treatment team may be willing to discharge them prior to reaching target weight, expecting the patients will continue to gain weight in less restrictive levels of care (i.e., outpatient). Future studies should examine whether personalizing treatment length based on early response in hospitalized patients with ARFID is predictive of outcome.

The slower rate of weight gain observed in the ARFID group may reflect that those with ARFID had more difficulty with meal completion compared to patients with AN, which may be due to the fact that the nutritional rehabilitation protocol was designed for patients with AN. Similar to other ED treatment programs [13] and due to a lack of existing ARFID treatment guidelines, this treatment center uses an AN treatment protocol for patients with ARFID. Programs that adapted an AN treatment protocol to treat ARFID reported that ARFID patients required a more gradual exposure to non-preferred foods than those with AN [25], these modifications were not part of the current treatment protocol. Additionally, nasogastric (NG) feeds are commonly used for nutritional rehabilitation in ARFID, with one study reporting higher use of NG tubes in patents with ARFID compared to AN [30]. However, our program does not employ NG feeds for any patients. Although this study does not have the data on ARFID subtype, prior literature suggests that hospitalized patients with ARFID are more likely to suffer from fear of aversive consequences (e.g., choking; [16]. Existing protocols may need to be adapted to meet the unique needs of individuals with ARFID, which would depend on the disorder subtype (e.g., sensory sensitivities, lack of interest in food, or fear of aversive consequences). For example, adding an exposure intervention to specifically target patients' fears may be beneficial. Additionally, individualized treatment protocols may be needed for patients with ARFID to address life-long dietary restriction, improve efficacy, and Page 7 of 9

prevent relapse. Nutritional rehabilitation for patients with ARFID should be further studied to inform development of specialized treatment protocols for this population.

Self-reported symptomatology

As expected, adults with ARFID endorsed lower levels of drive for thinness, bulimia, and body dissatisfaction and higher levels of normative eating and body image self-efficacy. The normative eating and body image selfefficacy subscales of the EDRSQ assess difficulties with eating and body image related to fear of fatness specifically [21], which would explain relatively lower scores in the ARFID population. Adults with ARFID reported having higher desired BMI than patients with AN. Notable, unlike in the AN group, desired BMI was within the normal range, consistent with absence of fear of fatness in ARFID. Consistent with prior research [4, 8], adults with ARFID had lower state anxiety and depression than those with AN, but groups did not differ in trait anxiety. It is possible that patients with AN have higher state anxiety at admission due to fear of gaining weight during their hospitalization, which would not be present in patients with ARFID.

Adults with ARFID and AN did not differ on openness to experience, extroversion, conscientiousness, or agreeableness, although we only had enough power to detect a large effect size. However, adults with ARFID reported significantly lower neuroticism. Neuroticism is a disposition to experience negative affect and is the core dimension of internalizing psychopathology, which includes anxiety, depressive, and weight and shape-related eating disorders [10]. Lower neuroticism in adults with ARFID than in AN may highlight that these disorders are associated with different dimensions of psychopathology. AN is part of an internalizing disorders spectrum characterized by negative emotions [17]. Although ARFID has not been classified within dimensional models of psychopathology, some consider it to be similar to neurodevelopmental disorders, such as autism spectrum disorder and attention deficit hyperactivity disorder, with which ARFID is highly comorbid [26]. It is possible that different ARFID subtypes (i.e., sensory difficulties, fear of aversive consequences, lack of interest in food) may have different comorbidities and associations with psychopathology symptoms, which future research should investigate.

Limitations and conclusions

Several limitations should be noted. First, our sample size for the self-reported symptomatology analyses was small, and these results should be viewed as preliminary.

Second, we did not include measures of ARFID-specific psychopathology in this study. Future studies should include such assessments in addition to more traditional eating disorder weight and shape-related measures. Third, all analyses are correlational, and no causal relationships can be implied. Fourth, the samples were primarily Non-Hispanic Whites and the results may not generalize to minority populations. Fifth, as these were treatment-seeking underweight adults with ARFID and AN results may not generalize to other ARFID populations. Sixth, we do not have data on ARFID subtypes (i.e., sensory difficulties, fear of aversive consequences, lack of interest in food), and findings may be affected by one subtype being more prevalent than the others. Sixth, we do not have follow-up data and are not able to draw conclusions regarding treatment outcome beyond discharge from inpatient treatment. Seventh, our sample is primarily female and the results may not generalize to male patients with ARFID. Finally, since length of illness is associated with ED severity, it may be important to match participants with ARFID and AN on this variable in addition to age and sex. Unfortunately, data on length of illness was not available for the current study.

This study adds to limited literature on presentations of ARFID in adults, specifically focusing on an underweight hospitalized population. Despite being closer to target weight at admission and reporting less comorbid psychopathology, adults with ARFID had less favorable weight-related outcomes during inpatient hospitalization, but long-term outcomes are unclear. More research is needed to better understand adult ARFID presentations to improve existing interventions.

Acknowledgements

Not applicable.

Author contributions

IV was responsible for conceptualization, methodology, formal analyses, data curation, and original writing. EW was responsible for original writing and visualization. MM was responsible for original writing. CC was responsible for review and editing. AG was responsible for conceptualization, review, and editing.

Funding

This work was supported in part by the Stephen and Jean Robinson Fund. The Fund was not involved in the content or writing of this article.

Availability of data and materials

The data that support the findings of this study are available on request from the corresponding author. The data are not publicly available due to privacy or ethical restrictions.

Declarations

Ethics approval and consent to participate

This study was approved by the Johns Hopkins School of Medicine Institutional Review Board. Informed consent was obtained from participants when applicable.

Consent for publication

Not applicable.

Competing interests

Authors have no competing interests to report.

Received: 26 May 2023 Accepted: 12 October 2023 Published online: 20 November 2023

References

- American Psychiatric Association. Diagnostic and Statistical Manual of Mental Disorders, 5 ed;2013.
- Beck AT, Steer RA, Brown GK. Manual for the Beck depression inventory-II. Psychological Corporation; 1996
- Becker, K. R., Keshishian, A. C., Liebman, R. E., Coniglio, K. A., Wang, S. B., Franko, D. L., Eddy, K. T., & Thomas, J. J. (2019). Impact of expanded diagnostic criteria for avoidant/restrictive food intake disorder on clinical comparisons with anorexia nervosa. *The International journal of eating disorders*, 52(3), 230–238. https://doi.org/10.1002/eat.22988
- Cañas L, Palma C, Molano AM, Domene L, Carulla-Roig M, Cecilia-Costa R, Dolz M, Serrano-Troncoso E. Avoidant/restrictive food intake disorder: psychopathological similarities and differences in comparison to anorexia nervosa and the general population. Eur Eat Disord Rev. 2021;29(2):245– 56. https://doi.org/10.1002/erv.2815.
- Costa PT, McCrae RR. The five-factor model of personality and its relevance to personality disorders. J Pers Disord. 1992;6(4):343–59. https:// doi.org/10.1521/pedi.1992.6.4.343.
- Faul F, Erdfelder E, Lang A-G, Buchner A. G*Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. Behav Res Methods. 2007;39(2):175–91. https://doi.org/10.3758/ BF03193146.
- First MB, Williams JBW, Karg RS, Spitzer RL. Structured clinical interview for DSM-5 disorders, clinician version (SCID-5-CV) (5th ed.). American Psychiatric Association;2016.
- Fjeldstad M, Kvist T, Sjögren M. Weight gain in adults with avoidant/ restrictive food intake disorder compared to restrictive anorexia nervosa—pilot findings from a longitudinal study. Nutrients. 2021;13(3):871. https://doi.org/10.3390/nu13030871.
- Garner DM, Olmstedt MP, Polivy J. Development and validation of a multidimensional Eating Disorder Inventory for anorexia and bulimia. Int J Eat Disord. 1983;2:14–34.
- Griffith JW, Zinbarg RE, Craske MG, Mineka S, Rose RD, Waters AM, Sutton JM. Neuroticism as a common dimension in the internalizing disorders. Psychol Med. 2010;40(7):1125–36. https://doi.org/10.1017/S003329170 9991449.
- Guarda AS, Cooper M, Pletch A, Laddaran L, Redgrave GW, Schreyer CC. Acceptability and tolerability of a meal-based, rapid refeeding, behavioral weight restoration protocol for anorexia nervosa. Int J Eat Disord. 2020;53(12):2032–7. https://doi.org/10.1002/eat.23386.
- Guillén-Riquelme A, Buela-Casal G. Metaanálisis de comparación de grupos y metaanálisis de generalización de la fiabilidad del cuestionario State-Trait Anxiety Inventory (STAI). Rev Esp Salud Publica. 2014;88(1):101–12. https://doi.org/10.4321/S1135-57272014000100007.
- Guss CE, Richmond TK, Forman S. A survey of physician practices on the inpatient medical stabilization of patients with avoidant/restrictive food intake disorder. J Eat Disord. 2018;6(1):22. https://doi.org/10.1186/ s40337-018-0212-4.
- Hilbert A, Zenger M, Eichler J, Brähler E. Psychometric evaluation of the Eating Disorders in Youth-Questionnaire when used in adults: prevalence estimates for symptoms of avoidant/restrictive food intake disorder and population norms. Int J Eat Disord. 2021;54(3):399–408. https://doi.org/ 10.1002/eat.23424.
- Kambanis PE, Harshman SG, Kuhnle MC, Kahn DL, Dreier MJ, Hauser K, Slattery M, Becker KR, Breithaupt L, Misra M, Micali N, Lawson EA, Eddy KT, Thomas JJ. Differential comorbidity profiles in avoidant/restrictive food intake disorder and anorexia nervosa: Does age play a role? Int J Eat Disord. 2022;55(10):1397–403. https://doi.org/10.1002/eat.23777.

- Katzman DK, Guimond T, Spettigue W, Agostino H, Couturier J, Norris ML. Classification of children and adolescents with avoidant/restrictive food intake disorder. Pediatrics. 2022;150(3): e2022057494. https://doi.org/10. 1542/peds.2022-057494.
- Kotov R, Krueger RF, Watson D, Achenbach TM, Althoff RR, Bagby RM, Brown TA, Carpenter WT, Caspi A, Clark LA, Eaton NR, Forbes MK, Forbush KT, Goldberg D, Hasin D, Hyman SE, Ivanova MY, Lynam DR, Markon K, Zimmerman M. The hierarchical taxonomy of psychopathology (HiTOP): a dimensional alternative to traditional nosologies. J Abnorm Psychol. 2017;126(4):454–77. https://doi.org/10.1037/abn0000258.
- Kurotori I, Shioda K, Abe T, Kato R, Ishikawa S, Suda S. An inpatient observational study: characteristics and outcomes of avoidant/restrictive food intake disorder (ARFID) in children and adolescents in Japan. Neuropsychiatr Dis Treat. 2019;15:3313–21. https://doi.org/10.2147/NDT.S218354.
- Makhzoumi SH, Schreyer CC, Hansen JL, Laddaran LA, Redgrave GW, Guarda AS. Hospital course of underweight youth with ARFID treated with a meal-based behavioral protocol in an inpatient-partial hospitalization program for eating disorders. Int J Eat Disord. 2019;52(4):428–34. https://doi.org/10.1002/eat.23049.
- Manwaring JL, Blalock DV, Rienecke RD, Le Grange D, Mehler PS (2023) A descriptive study of treatment-seeking adults with avoidant/restrictive food intake disorder at residential and inpatient levels of care. Eating Disord 1–16. https://doi.org/10.1080/10640266.2023.2241266
- Marinilli Pinto A, Guarda A, Heinberg LJ, Diclemente CC. Development of the eating disorder recovery self-efficacy questionnaire. Int J Eat Disord. 2006;39(5):376–84. https://doi.org/10.1002/eat.20256.
- Nakai Y, Nin K, Noma S, Hamagaki S, Takagi R, Teramukai S, Wonderlich SA. Clinical presentation and outcome of avoidant/restrictive food intake disorder in a Japanese sample. Eat Behav. 2017;24:49–53. https://doi.org/ 10.1016/j.eatbeh.2016.12.004.
- Oei TPS, Evans L, Crook GM. Utility and validity of the STAI with anxiety disorder patients. Br J Clin Psychol. 1990;29(4):429–32. https://doi.org/10. 1111/j.2044-8260.1990.tb00906.x.
- Pinto AM, Heinberg LJ, Coughlin JW, Fava JL, Guarda AS. The eating disorder recovery self-efficacy questionnaire (EDRSQ): change with treatment and prediction of outcome. Eat Behav. 2008;9(2):143–53. https://doi.org/ 10.1016/j.eatbeh.2007.07.001.
- Rienecke RD, Drayton A, Richmond RL, Mammel KA. Adapting treatment in an eating disorder program to meet the needs of patients with ARFID: Three case reports. Clin Child Psychol Psychiatry. 2020;25(2):293–303. https://doi.org/10.1177/1359104519864129.
- Sanchez-Cerezo J, Nagularaj L, Gledhill J, Nicholls D. What do we know about the epidemiology of avoidant/restrictive food intake disorder in children and adolescents? A systematic review of the literature. Eur Eat Disord Rev: J Eat Disord Assoc. 2023;31(2):226–46. https://doi.org/10. 1002/erv.2964.
- Spielberger CD. Manual for the State-Trait Anxiety Inventory (STAI Form Y). Consulting Psychologists Palo Alto: Consulting Psychologists Press Inc; 1983.
- Steer RA, Ball R, Ranieri WF, Beck AT. Further evidence for the construct validity of the beck depression inventory-II with psychiatric outpatients. Psychol Rep. 1997;80(2):443–6. https://doi.org/10.2466/pr0.1997.80.2.443.
- Steinberg, D. M., Perry, T. R., Freestone, D., Hellner, M., Baker, J. H., & Bohon, C. (2023). Evaluating differences in setting expected body weight for children and adolescents in eating disorder treatment. *The International journal of eating disorders*, *56*(3), 595–603. https://doi.org/10.1002/eat. 23868
- Strandjord SE, Sieke EH, Richmond M, Rome ES. Avoidant/restrictive food intake disorder: illness and hospital course in patients hospitalized for nutritional insufficiency. J Adolesc Health. 2015;57(6):673–8. https://doi. org/10.1016/j.jadohealth.2015.08.003.
- Tasca GA, Demidenko N, Krysanski V, Bissada H, Illing V, Gick M, Weekes K, Balfour L. Personality dimensions among women with an eating disorder: towards reconceptualizing DSM. Eur Eat Disord Rev. 2009;17(4):281–9. https://doi.org/10.1002/erv.938.
- Thiel A, Paul T. Test-retest reliability of the Eating Disorder Inventory 2. J Psychosom Res. 2006;61:567–9. https://doi.org/10.1016/j.jpsychores.2006. 02.015.
- Thomas JJ, Becker KR, Breithaupt L, Murray HB, Jo JH, Kuhnle MC, Dreier MJ, Harshman S, Kahn DL, Hauser K, Slattery M, Misra M, Lawson EA, Eddy KT. Cognitive-behavioral therapy for adults with avoidant/restrictive food

 Zickgraf HF, Franklin ME, Rozin P. Adult picky eaters with symptoms of avoidant/restrictive food intake disorder: comparable distress and comorbidity but different eating behaviors compared to those with disordered eating symptoms. J Eat Disord. 2016;4:26. https://doi.org/10.1186/ s40337-016-0110-6.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

